Author’s response to reviews

Title: Clinically Significant Prostate Cancer Diagnosed Using a Urinary Molecular Biomarker-based Risk Score: Two Case Reports

Authors:

Pieter Minnee (P.Minnee@llz.nl)

Daphne Hessels (daphne.hessels@mdxhealth.com)

Jack Schalken (Jack.Schalken@radboudumc.nl)

Wim Van Criekinge (wim.vancriekinge@gmail.com)

Version: 2 Date: 20 Nov 2019

Author’s response to reviews:

Dear Editor,

Thank you for considering our manuscript for publication. Based on your comments:

1. We have added the Discussion and Conclusions section.

2. We have supplied the consent statement.

3. We have provided a single clean copy of the manuscript

Kind regards,

Prof Dr Wim Van Criekinge